Author's response to reviews

Title: Efficiency of two-phase methods with focus on a planned population-based case-control study on air pollution and stroke

Authors:

Anna Oudin (anna.oudin@med.lu.se)
Jonas Bjork (jonas.bjork@skane.se)
Ulf Stromberg (ulf.stromberg@med.lu.se)

Version: 8 Date: 6 November 2007

Author's response to reviews:

Dear editors,
We've saved the files in the rtf-format, to avoid endnote complications.
I hope it was sucessful.
Sincerely,
Anna Oudin